Author's response to reviews

Title: Effect of sunlight exposure on cognitive function among depressed and non-depressed participants: a REGARDS cross-sectional study

Authors:

Shia T Kent (shia@uab.edu)
Leslie A McClure (lmcclure@uab.edu)
William L Crosson (bill.crosson@nasa.gov)
Donna K Arnett (Arnett@ms.soph.uab.edu)
Virginia G Wadley (vwadley@uab.edu)
Nalini Sathiakumar (NSathiakumar@ms.soph.uab.edu)

Version: 8 Date: 19 June 2009

Author's response to reviews:

I have changed only a few typos regarding spelling and grammar. No content or other fundamental changes.